\begin{thebibliography}{10}

\bibitem{Abramsky87}
S.~Abramsky and C.~L.~(eds) Hankin.
\newblock {\em {Abstract Interpretation of Declarative Languages}}.
\newblock Ellis-Horwood, 1987.

\bibitem{Ariola89}
Z.~Ariola and Arvind.
\newblock {P-TAC: A Parallel Intermediate Language}.
\newblock In {\em Proceedings of the ACM Conference on Functional Programming
  Languages and Computer Architecture, London, UK}, pages 230--242. ACM,
  September 1989.

\bibitem{Ariola91}
Z.~Ariola and Arvind.
\newblock {Compilation of Id}.
\newblock In {\em Proc. of the Fourth Workshop on Languages and Compilers for
  Parallel Computing (LNCS 589)}. Springer-Verlag, August 1991.

\bibitem{Bloss88}
A.~Bloss.
\newblock {\em {Path Analysis and the Optimization of Non-strict Functional
  Languages}}.
\newblock PhD thesis, Dept. of Computer Science, Yale University, May 1989.

\bibitem{Bloss87}
A.~Bloss and P.~Hudak.
\newblock {Path Semantics}.
\newblock In {\em Mathematical Foundations of Programming Language Semantics
  (LNCS 298)}. Springer-Verlag, April 1987.

\bibitem{Burn86}
J.~Burn, C.~Hankin, and S.~Abramsky.
\newblock {Strictness Analysis for Higher-order Functions}.
\newblock {\em Science of Computer Programming}, 9, 1986.

\bibitem{Clack85}
C.~Clack and S.~L. Peyton-Jones.
\newblock {Strictness Analysis -- A Practical Approach}.
\newblock In {\em Proceedings of ACM Conference of Functional Programming
  Languages and Computer Architecture}. ACM, September 1985.

\bibitem{Coorg94}
S.~R. Coorg.
\newblock {Partitioning Non-strict Languages for Multi-threaded Code
  Generation}.
\newblock Masters Thesis, EECS, MIT, May 1994.

\bibitem{Culler93}
D.~E. Culler, S.~C. Goldstein, K.~E. Schauser, and T.~von Eicken.
\newblock {TAM --- A Compiler Controlled Threaded Abstract Machine.}
\newblock {\em {Journal of Parallel and Distributed Computing}},
  18(4):{347--370}, July 1993.

\bibitem{Garey79}
M.~R. Garey and D.~S. Johnson.
\newblock {\em {Computers and Intractability: A Guide to the Theory of
  NP-Completeness}}.
\newblock W.H. Freeman, San Francisco, CA, 1979.

\bibitem{Henderson80}
P.~Henderson.
\newblock {\em {Functional Programming: Applications and Implementation}}.
\newblock Prentice Hall, Englewood Cliffs, NJ, 1980.

\bibitem{Hoch91}
J.~E. Hoch, D.~M. Davenprot, V.~G. Grafe, and K.~M. Steele.
\newblock {Compile-time Partitioning of a Non-strict Language into Sequential
  Threads}.
\newblock In {\em Proc. Symp. on Parallel and Distributed Processing}, Dec
  1991.

\bibitem{Haskell90}
P.~Hudak and P.~Wadler~(editors).
\newblock {Report on the Programming Language {H}askell, A Non-strict Purely
  Functional Language ({V}ersion 1.0)}.
\newblock Technical Report YALEU/DCS/RR777, Yale University, Department of
  Computer Science, April 1990.

\bibitem{Hudak86}
P.~Hudak and J.~Young.
\newblock {Higher-order Strictness Analysis of the Untyped Lambda-Calculus}.
\newblock In {\em Proc. 12th ACM Symposium on Principles of Programming
  Languages}, pages 97--109. ACM, Jan 1986.

\bibitem{Ianucci88}
R.~A. Ianucci.
\newblock {\em {A Dataflow/von Neumann Hybrid Architecture}}.
\newblock PhD thesis, Dept. of EECS, MIT, May 1988.
\newblock (Also MIT LCS TR-418).

\bibitem{Mycroft80}
A.~Mycroft.
\newblock {The Theory and Practice of Transforming Call-by-need into
  Call-by-value}.
\newblock In {\em {International Symposium on Programming (LNCS 83)}}.
  Springer-Verlag, April 1980.

\bibitem{Nikhil90}
R.~S. Nikhil.
\newblock {Id Language Reference Manual Version 90.1}.
\newblock Technical Report CSG Memo 284-2, MIT Laboratory for Computer Science,
  545 Technology Square, Cambridge, MA 02139, July 15 1991.

\bibitem{Nikhil92}
R.S. Nikhil, G.M. Papadopoulos, and Arvind.
\newblock {*T: A Multithreaded Massively Parallel Architecture}.
\newblock In {\em Proc. 19th Annual Intl. Symp. on Computer Architecture}, May
  1992.

\bibitem{Schauser94b}
K.~E. Schauser.
\newblock {\em {Compiling Lenient Languages for Parallel Asynchronous
  Execution}}.
\newblock PhD thesis, Computer Science Div., Univ. of California at Berkeley.,
  May 1994.

\bibitem{Schauser94a}
K.~E. Schauser, D.~Culler, and S.~C. Goldstein.
\newblock {Separation Constraint Partitioning -- A New Algorithm for
  Partitioning Non-strict Programs into Sequential Threads}.
\newblock In {\em Proc. ACM Conference on Principles of Programming Languages
  (POPL)}, January 1995.

\bibitem{Schauser91}
K.~E. Schauser, D.~Culler, and von Eicken~T.
\newblock {Compiler-controlled Multithreading for Lenient Parallel Languages}.
\newblock In {\em Proc. Conf. on Functional Programming Languages and Computer
  Architecture}, August 1991.

\bibitem{Schmidt}
D.~A. Schmidt.
\newblock {\em {Denotational Semantics -- A Methodology for Language
  Development}}.
\newblock Allyn and Bacon, Boston, MA, 1986.

\bibitem{Traub86}
K.~R. Traub.
\newblock {A Compiler for the MIT Tagged-Token Dataflow Architecture}.
\newblock Technical Report TR-370, MIT Lab. for Computer Science, August 1986.
\newblock (MS Thesis, Dept. of EECS, MIT).

\bibitem{Traub89}
K.~R. Traub.
\newblock {Compilation as Partitioning: A New Approach to Compile Non-strict
  Functional Languages}.
\newblock In {\em Proc. of the ACM Conf. on Functional Programming and Computer
  Architecture}, pages 75--88. ACM, 1989.

\bibitem{Traub88}
K.~R. Traub.
\newblock {\em {Implementation of Non-strict Functional Programming
  Languages}}.
\newblock MIT Press, Cambridge, MA, 1991.

\bibitem{Traub92}
K.~R. Traub, D.~E. Culler, and K.~E. Schauser.
\newblock {Global Analysis for Partitioning Non-strict Programs into Sequential
  Threads}.
\newblock In {\em Proc. of the ACM Conf. on LISP and Functional Programming},
  June 1992.

\bibitem{Wadler87}
P.~Wadler.
\newblock {Strictness Analysis on Non-Flat Domains}.
\newblock In {\em \cite{Abramsky87}}.

\end{thebibliography}

\section{Notation, Syntax and Semantics}
\label{section:syntax}
We design our algorithms on a simple {\em kernel} language,
SP-TAC (Simple P-TAC) language based on the intermediate language P-TAC 
(Parallel Three-Address Code) \cite{Ariola89} that is used during
the compilation of Id. SP-TAC is 
a simple first order functional language, with constants, variables,
primitive operators, conditionals, user defined functions and
recursive {\em let}-blocks. It is a restricted form of P-TAC in 
several ways: there are no higher order functions, 
structured data types (like lists), or side-effects. 
Though SP-TAC is a very restrictive language to write programs in,
it is a non-strict language and it brings out the issues in
compiling non-strictness. Also,
techniques developed for partitioning SP-TAC are directly applicable
to more expressive languages.

\subsection{Abstract Syntax of SP-TAC}
\begin{myfig}{figure:syntax}{The Syntax of a Kernel Language}
\centering
\begin{tabular}{lcl}
$c$ & $\in$  & Constants\\
 $x,y$ & $\in$ &  Variables \\
 $p$  & $\in$ & Primitive operators \\
 $f$ & $\in$  &  User defined functions \\
 $se$ & $\in$ &  Simple expressions \\
 $e$ & $\in$ & Expressions\\
$st$ & $\in$ & Statements \\
$b$ & $\in$ & Blocks \\
$pr$ & $\in$ & Programs \\\\
$c$ & $::=$ & ${\bf \, true \:|\:false \:| 1\: |\: 2\:|\ldots}$ \\
$p$ & $::=$ & ${\bf\: + \:|\:-\:|\:and\:|\ldots}$\\
$se$ & $::=$ & $c\:|\:x$\\
$e$ & $::=$ & $se \: |\: b \:|\:p(se_{1},se_{2},\ldots,se_{n})\: |$\\
& & $f(se_{1},se_{2},\ldots,se_{n}) \: | \: {\bf if}\:se\:{\bf then}\:b_{1}\:{\bf else}\:b_{2}$\\
$st$ & $::=$ & $\:(y_1,y_2,\ldots,y_n)\:=\:e$\\
$b$ & $::=$ & $\{\{st\}^*\:{\bf in}\:(se_1,se_2,\ldots,se_n)\}$\\
$pr$ & $::=$ & $\{f_{i}(x_{1},x_{2},\ldots,x_{n})\:=\:b_{i}\}$
\end{tabular}
\end{myfig}
Figure~\ref{figure:syntax} shows the abstract syntax of SP-TAC.
We make the following assumptions about the language.
\begin{itemize}
\item There is no currying of user defined functions. This ensures
that we are dealing with a first order language.
\item All primitive operators are strict. That is, they require
all their inputs to produce their output value.
\item All local and bound variables are $\alpha$-renamed so that they
are defined uniquely in the program.
\end{itemize}
For notational convenience, examples in this paper are given in Id. 
It is straightforward to translate them into programs in the kernel language
\cite{Ariola91}.

\subsection{Representing SP-TAC using Graphs}
In this paper, we also use a {\em graph} representation of a block
of SP-TAC. This representation (called a {\em dataflow graph}) is similar to 
the one used in the Id compiler to perform various
optimizations \cite{Traub86}. 
The language syntax
shown in the previous section is essentially a textual representation of these
dataflow graphs. 

\begin{definition} [Dataflow Graph]
Given a block $b$, the dataflow graph $G(b)$ corresponding to $b$ is 
a directed graph. The nodes $N(b)$ in the graph consist of {\em input nodes}
-- 
corresponding to the free variables of the block, {\em output nodes} --
corresponding to the values returned by the block, and {\em operator
nodes} -- corresponding to the various operators
specified in the block.
The edges $E(b)$ of this graph correspond to the flow of values in the block. 
\end{definition}
In this paper, we treat the graph representation and the textual 
representation
as equivalent. For example, when we refer to {\em nodes} in a block, we
actually refer to the nodes in the dataflow graph corresponding to 
the block. Figure~\ref{figure:dataflow-graph} shows the dataflow graph
corresponding to the functions defined in examples 1 and 2 in
section~\ref{section:introduction}.
\begin{myfig}{figure:dataflow-graph}{Figures (a)
and (b) show examples of dataflow graphs.
Input and output nodes are denoted
by circles and constant nodes by triangles. Rectangular nodes correspond
to the various operators in the function.}
\illustration{fig10}[\hsize]
\vspace{-4ex}
\end{myfig}

\subsection{Semantics}
\begin{myfig}{figure:semantic-equations}{Semantic Equations of SP-TAC.}
\centering
$\begin{array}{lcl}
{\cal E}_c\ld n \rd & = & n, \mbox{integer $n$} \\
{\cal E}_c\ld true \rd & = & true \\		
{\cal E}_c \ld false \rd & = & false \\\\
{\cal E}_k\ld + \rd & = & \lambda(x,y).(Int?(x)\: and \: Int?(y)) \\
& & \rightarrow  x + y, error\\\\
{\cal E}_s \ld c \rd \: bve & = & {\cal E}_c \ld c \rd\\
{\cal E}_s \ld x \rd \: bve & = & bve \ld x \rd\\\\
{\cal E} \ld se \rd \: bve \: fenv  & = & {\cal E}_s \ld se \rd \: bve\\
\multicolumn{3}{l}{{\cal E} \ld p(se_1,se_2,\ldots,se_n) \rd \: bve \: fenv =}\\
\multicolumn{3}{l}{\hspace{0.25in} {\cal E}_k \ld p \rd \: ({\cal E}_s \ld se_1 \rd \: bve, \ldots,{\cal E}_s \ld se_n \rd \: bve)}\\
\multicolumn{3}{l}{{\cal E} \ld f(se_1,se_2,\ldots,se_n) \rd \: bve \: fenv =}\\
\multicolumn{3}{l}{\hspace{0.25in} fenv \ld f \rd
\: ({\cal E}_s \ld se_1 \rd \: bve, \ldots,{\cal E}_s \ld se_n \rd \: bve)}\\
\multicolumn{3}{l}{{\cal E} \ld {\bf if}\: se \: {\bf then} \: b_1 {\bf else} \: b_2 \rd \: bve \: fenv = }\\
\multicolumn{3}{l}{\hspace{0.25in} ({\cal E}_s\ld se \rd \: bve) \rightarrow ({\cal E}_b\ld b_1 \rd \: bve \: fenv),({\cal E}_b \ld b_2 \rd \: bve \: fenv)}\\
{\cal E} \ld b \rd \: bve \: fenv  & = & {\cal E}_b \ld b \rd \: bve \: fenv\\
\\
\multicolumn{3}{l}{{\cal E}_b\ld \{\{(y_{i1},\ldots,y_{im})=e_i\}^*\: {\bf in}\: (se_1,\ldots,se_n) \}\rd  \: bve \: fenv = }\\
\multicolumn{3}{l}{\mbox{\hspace{0.25in} \bf letrec}}\\
\multicolumn{3}{l}{\hspace{0.25in} newbve=[y_{ij} \mapsto ({\cal E}\ld e_i \rd \: (newbve.bve) \: fenv)\downarrow j]}\\
\multicolumn{3}{l}{\mbox{\hspace{0.25in} \bf in} \: ({\cal E}_s\ld se_1\rd  \: newbve,\ldots,{\cal E}_s\ld se_n\rd  \: newbve)}\\
\\
\multicolumn{3}{l}{{\cal E}_p\ld \{f_i(x_1,x_2,\ldots,x_n)=b_i\}^*\rd = fenv, \mbox{\bf whererec}}\\
\multicolumn{3}{l}{\hspace{0.25in}fenv = [f_i \mapsto (\lambda(y_1,y_2,\ldots,y_n).{\cal E}_b\ld b_i \rd \: [x_i \mapsto y_i]\: fenv)]}
\end{array}$
\end{myfig}

Our framework is based on denotational semantics. We  chose to do this 
because techniques for designing and
proving analyses based on abstract interpretation are 
well developed in this framework 
and these techniques handle recursive function definitions in a natural manner.
The semantics of SP-TAC itself is straightforward as it is a simple first
order language. It is based on the semantics given in \cite{Bloss88}
and specifies a non-strict semantics of the language.

Denotational semantics \cite{Schmidt} is given by defining domains 
(complete partially ordered sets)
and constructing a mapping from entities in the language
to elements of these domains. 
For SP-TAC, we need to define the following domains. 
We have the standard domains $Int$ and $Bool$ for integers and booleans. 
These domains contain an element $\bot$ that stands for the ``undefined''
value. To model multiple values, we use {\em product} domains
\cite{Schmidt}. We use $D^k$ to denote a domain formed by the k-product $D \times D \times \ldots \times D$. 
The infix operator $\downarrow$ is used to select any component of 
a value in $D^k$.

In addition to these domains, we have domains of functions that
map (multiple) values to (multiple) values. There are also domains
corresponding to variable environments and function environments.
These domains map variables to values and function names to functions
respectively. 
The formal definition of the domains we use in our
semantics are:
\begin{center}
\begin{tabular}{lr}
$Int$ & Integers \\
$Bool$ & Booleans \\
$Val = Int + Bool$& Values\\
$Fun = \bigcup_{j,k \geq 1} (Val^j \rightarrow Val^k)$&First order functions\\
$D=Val + \{error\}$& Denotable values\\
$D_T = \bigcup_{k \geq 1} Val^k + \{error\}$& Tuples of values\\
$Bve=V \rightarrow D$& Variable Environments\\
$Fenv=Fv \rightarrow Fun$&Function Environments\\
\end{tabular}
\end{center}
The various semantic functions are given below:
\begin{center}
\begin{tabular}{lr}
${\cal E}_c : Con \rightarrow Val$ & Constants\\
${\cal E}_k : Pf \rightarrow Fun$ & Primitive operators\\
${\cal E}_s : SE \rightarrow Bve \rightarrow D$ & Simple Exps\\
${\cal E}_b : Bo \rightarrow Bve \rightarrow Fenv \rightarrow D_T$
& Blocks\\
${\cal E} : Exp \rightarrow Bve \rightarrow Fenv \rightarrow D_T$
& Expressions\\
${\cal E}_p : Prog \rightarrow Fenv$ & Programs
\end{tabular}
\end{center}

Now, we describe semantic equations that define the behavior
of these semantic functions. Here is some notation used in describing
our semantic equations:
\begin{itemize}
\item If $env$ and $env'$ are two environments, we use $env'.env$ to denote the 
environment $env''$ such that $env''\ld y \rd = env' \ld y \rd$ if 
$y \in Domain(env')$, $env'' \ld y \rd = env \ld y \rd$, otherwise. 
This is the way environments are linked in a statically scoped block structured
language.
\item We also make use of a case operator 
(denoted by $\rightarrow$). The meaning of $x \rightarrow y, z$ is 
that depending on whether $x$ is $true$ or $false$ either $y$ or $z$ is
the value of the entire case expression.
\end{itemize}

The semantic equations are given in figure~\ref{figure:semantic-equations}.
For clarity of the equations, we have omitted the projections and
injections between the various domains.
Given a program $pr$, we use ${\cal E}^{pr} \ld b \rd \: bve$ to denote ${\cal E} \ld b \rd
\: bve \: ({\cal E}_p \ld pr \rd)$.


